1	1	Wordcount: 12014 words
2 3 4	2	Stimulate hydropower output of mega cascade reservoirs using an
5 6 7	3	improved Kidney Algorithm
8 9 10	4	
11 12 13	5	Yanlai Zhou ^{1,2} , Shenglian Guo ^{1,*} , Chong-Yu Xu ^{2,*} , Fi-John Chang ³ , Hua Chen ¹ , Pan Liu ¹ , Bo
14 15	6	Ming ¹
16 17 18	7	¹ State Key Laboratory of Water Resources and Hydropower Engineering Science, Wuhan
19 20 21	8	University, Wuhan, 430072, China.
22 23 24	9	² Department of Geosciences, University of Oslo, P.O. Box 1047 Blindern, N-0316 Oslo,
24 25 26	10	Norway.
27 28 29	11	³ Department of Bioenvironmental Systems Engineering, National Taiwan University, Taipei,
30 31 32	12	10617, Taiwan.
33 34 35 36 37 38 90 41 42 44 45 47 48 90 512 55 57 58 601 612	13	1
62 63 64 65		1

Abstract

The evolutionary algorithms can solve reservoir operation with a fast convergence rate whereas the major impediments in handling the joint operation of mega cascade reservoirs easily trigger the technical bottlenecks, i.e. trapping into a local optimum, instability and loss of good solutions. This study proposes a methodology that fuses three auxiliary strategies into the Kidney Algorithm (KA) to optimize the hydropower output for conquering the bottlenecks in the KA concerning the joint operation of six mega cascade reservoirs located in the Jin-Sha River basin in China. The proposed theme would contribute to the application of the state-of-the-art evolutionary algorithms in boosting the cleaner hydropower production of mega cascade reservoirs. The three auxiliary strategies are that: firstly, the exploration and exploitation strategy is employed to stimulate the movement of solutions to surmount technical drawback of trapping into a local optimum; secondly, the adaptive strategy is used to automatically adjust algorithm parameter values to overcome the instability problem; lastly, the elitism strategy is introduced to preserve the best solution at every epoch to avoid the loss of good solutions. Our methodology, without expanding or upgrading hydraulic infrastructures, can increase the hydropower production of the six mega cascade reservoirs by 7.8 %, as compared with the standard operation policy. The hydropower production can reach 4.8 billion kW[·]h/year, which can decrease 3.77 billion kg/year in CO₂ emission, and bring 217.44 million USD/year in hydropower benefits. The improved KA can considerably increase the reliability and resilience of hydropower output as well as largely decrease the vulnerability of hydropower output. The results suggest that our methodology can stimulate hydropower output to yield more benefits regarding cleaner production, carbon emission reduction and sustainability.

Keywords: Hydropower production; Cascade reservoirs; Optimal operation; Water-Energy
 nexus; Artificial Intelligence (AI); Jin-Sha River

1. Introduction

Renewable energy spreads to a growing number of developing and emerging economies. In some areas, renewable energy has become a pivotal electricity source due to the rapid growth in the population under urbanization. The ongoing growth in magnitude and geographical expansion of renewable power capacity are driven by the continuing decline in price for renewable energy technologies, by raising power demand in some countries and by targeting renewable energy support mechanisms. Nowadays, most new renewable energy power plants are installed in developing countries, especially in China, which is the largest developer over the past eight years. By the end of 2016, the top regions or countries for total installed renewable energy capacity are China, Europe, USA, India and Japan (Figure 1 (a)). In 2016, the renewable energy production estimated to reach 30 % (2016.8 GW) of the world's generation capacity. This amount is enough to provide 24.5 % of global actual energy consumption. Among the renewable energy sources, hydropower has a low mean power generation cost and high generation stability (Global Status Report of renewable energy, 2017). Globally, hydropower provides 16.6 % global energy consumption (Figure 1 (b)), and this number exceeds 20 % in China (REN21, 2017). The hydropower will continue to grow (from 13 % in 2000 to 19 % in 2016) to compensate for the decline in thermal power production (from 85 % in 2000 to 73 % in 2016). Compared with other renewable energy sources, hydropower is flexible in electricity generation and supply, and hence hydropower



Renewable energy	Global	EU-28	China	USA	India	Japan		
technology		GW						
≈ Hydropower	1096	127	332	80	47	23		
Wind power	487	154	169	82	29	3.2		
🔆 Solar PV	303	106	77	41	9.1	43		
Bio-power	112	37	12	16.8	8.3	4.1		
Others	18.8	3.5	0.4	5.3	0.2	0.5		
b.								
Non-renewable power 75.5 %			Hydro	power	•	Wind power 4.0 %		
			16.6 %			Bio-power 2.0 %		
	Renewable	power			•	Solar PV		

1.5 %

Others 0.4 %

Figure 1 Global renewable energy capacity and production in 2016. a. Global renewable
energy capacity of top regions/countries. b. Renewable energy share of global energy
production.

24.5 %

Notes: EU-28 consists of 28 European Countries. The data tracked 155 countries including Africa, Asia, Central America, the Caribbean, Eurasia, Europe, Middle East, North America, Oceania, South America, China, India and the United States, covering 96% of global GDP and representing 96% of global population. (Extracted from the REN21 Renewables Global Status Report, 2017).

- **68**

yields more social benefits for energy economy (He et al., 2018), energy safety (Cheng et al., 2018), carbon emission reduction (Hu et al., 2011; Dou, 2013) and non-fossil energy expansion (Feng et al., 2018a, b). Besides, many countries and regions are working to improve hydropower infrastructure, operation and market design to facilitate hydropower output (Ehteram et al., 2017; Singh and Singal, 2017). To raise cleaner production, our study is concentrated on probing into a joint operation of mega cascade reservoirs to lift synergy of water-energy nexus and significantly mitigate CO₂ emission with the use of Artificial Intelligence (AI)-based heuristic techniques.

Modernization and retrofitting of existing facilities continue to be a vital part of hydropower operations, including the implementation of advanced AI technologies and data analytics for digitally enhanced hydropower generation (Singh and Singal, 2017; Jha et al., 2017). In recent years, researchers are seeking to imitate nature by evolutionary algorithms because the designs and abilities of nature are tremendous (Fister et al., 2013; Molina et al., 2018), and therefore nature is the best trainer for technology. Since the two domains and fields (nature & technology) have a much stronger connection and similarity, easy mapping is possible from nature to technology in the real world. Evolutionary algorithms inspired by nature mechanisms and used as a branch of AI techniques for solving various optimization problems have evolved rapidly over the last few decades (Maarouf et al., 2015; Allawi et al., 2019). The evolutionary algorithms are derived from the activities of physical or biological systems in the natural world. Some examples of evolutionary algorithms in the literature are listed in Table 1. The Genetic Algorithm (GA) (Goldberg, 1989), Simulated Annealing (SA) (Johnson et al., 1989), Particle Swarm Optimization (PSO) (Kennedy and Eberhart, 1995),

Harmony Search (HS) (Geem et al., 2001), Ant Colony Optimization (ACO) (Bianchi et al., 2002), Honey Bee Optimization (HBO) (Pham et al., 2005), Intelligent Water Drops (IWD) (Hosseini, 2007), Cuckoo Search (CS) (Yang and Deb, 2009), Bat Algorithm (BA) (Yang, 2010a), Firefly Algorithm (FA) (Yang, 2010b), Black Hole (BH) (Hatamlou, 2013) and Kidney Algorithm (KA) (Jaddi et al., 2017) have been widely applied to optimizing hydropower stations (or cascade reservoirs) long term operation and short term operation as well as to renewable energy hybrid operation. For instance, Wang et al. (2018) proposed an effective procedure to strengthen the hydropower scheme by minimizing spillages in the cascade reservoirs short-term operation. Uen et al., (2018) developed a holistic scheme that integrated the long-term and short-term reservoir operation for improving the synergistic benefits of water-energy nexus. Ming et al. (2018a, b) fused the CS algorithm into dynamic programming to optimize the joint operation of large hydro-photovoltaic hybrid power plants. Shen et al. (2019) combined evolutionary algorithm and decision-making analysis to optimize the operation of interprovincial hydropower System. In comparison to the above mentioned evolutionary algorithms, the KA is introduced to optimize joint operation of mega cascade reservoirs on the grounds that: firstly, the KA was introduced by Jaddi et al. (2017) as a successful state-of-the-art optimization algorithm suitable for different engineering applications versus the other algorithms (Ekinci et al., 2018; Jaddi and Abdullah, 2018) in term of its computation speed, convergence, stability, and secondly, a review of the available literature indicates the KA has not been applied in mega cascade reservoirs operation. KA's application for the first time to a reservoir operation made by Ehteram et al. (2018a, b). To the best of our knowledge, although the KA can be used to solve the optimization of low

dimensional reservoir operation (e.g. one reservoir, 12 (months) decision variables and 48 (= 12 months * 4 constraints) physical constraints at monthly time scale in a year), but its reliability and practicality of solving the high dimensional cascade reservoirs operation has not been explored. The major difficulties in handling a large number of decision variables and constraints closely associated with the optimization of cascade reservoirs operation and non-convex objective function (Cheng et al., 2012). They easily trigger the technical drawbacks, i.e. trapping into a local optimum, loss of good solutions as well as the instability problem (or lack of robustness) in evolutionary algorithms. Consequently, it is imperative to conduct indepth research on the KA for enhancing its robustness of exploration and exploitation in solving the nonlinear non-convex objective function and high dimensional optimization operation of mega cascade reservoirs.

Table 1 Examples of evolutionary algorithms in the literature

1	Evolutionary algorithms	Imitation	References
2 3	Genetic Algorithm (GA)	Natural selection operator and genetic variation	Goldberg, 1989.
4	Simulated Annealing (SA)	Steel annealing process	Johnson et al., 1989.
5	Particle Swarm Optimization (PSO)	Swarm behavior	Kennedy and Eberhart, 1995.
6	Harmony Search (HS)	Finding the harmony in music	Geem et al., 2001.
7 8	Ant Colony Optimization (ACO)	Finding shortest path to the food sources of ants	Bianchi et al., 2002.
9	Honey Bee Optimization (HBO)	Food-foraging behavior of honey bee colonies	Pham et al., 2005.
0	Intelligent Water Drops (IWD)	Destination finding behavior of natural rivers	Hosseini, 2007.
1	Cuckoo Search (CS)	Reproduction behavior of the cuckoo	Yang and Deb, 2009.
∠ २	Bat Algorithm (BA)	Echolocation behavior of bat	Yang, 2010a.
4 5	Firefly Algorithm (FA)	Flashing light emitted by fireflies in the natural world	Yang, 2010b.
6	Black Hole (BH)	Black hole phenomenon	Hatamlou, 2013.
7	Kidney Algorithm (KA)	Kidney process in the human body	Jaddi et al., 2017.
R .			

The main objective of this study is to promote the application of the state-of-the-art evolutionary algorithms for improving the cleaner hydropower production of mega cascade reservoirs. The innovative nature of this study lies in fusing three auxiliary strategies into the KA to overcome its technical bottlenecks. The improved KA is applied for optimizing the

hydropower production of six mega cascade reservoirs. This is the first time that the KA is modified by using three auxiliary strategies and used to solve a complex joint operation of mega cascade reservoirs. The exploration is placed on two focuses. Firstly, the cascade reservoirs operation objective is defined as to maximize the hydropower generation, which a penalty function is added to the objective function to avoid violations of the guaranteed (or firm) power output. Secondly, an improved KA with three auxiliary strategies is employed to solve the optimization problem in a hierarchical structure. The auxiliary strategies consist of: for the movement operator and filtration operator, the exploration and exploitation strategy is introduced to stimulate the movement of solutions, and the adaptive strategy is used to adjust algorithm parameter values respectively. Before reaching the maximum epoch or stopping criterion, the elitism strategy is adopted for preserving the best solution in every epoch. The six mega cascade reservoirs located at the middle reach of Jin-Sha River in China are selected as a case study to assess the applicability as well as reliability of the proposed method.

This paper is organized into five sections. Section 2 introduces the study area and data. Section 3 describes the framework of the proposed method consisting of the joint operation model of mega cascade reservoirs, the standard KA and the improved KA. Section 4 presents results and discussion on the methods in the study case. Section 5 summarizes the results.

148 2. Study area and data

Effective management of hydropower stations is the key to the sustainability of our energy sources of tomorrow. China has greatly endeavoured to make transit-oriented development of renewable energy systems for fulfilling the pledge of carbon emission reduction and non-

fossil energy expansion to 20% by 2030 or earlier. The installed hydropower capacity of China reached 332 GW by the end of 2016, which was attributed to the fast development of hydropower resources and the intensive construction of power grids during the past three decades. Hydropower resources are concentrated mainly in south-western China while electricity loads occur mainly around the Yangtze River Delta and the Pearl River Delta. Being credited to the merits in nature, the Yangtze River basin possesses the largest water and hydropower resources in China. A total of 267 large reservoirs (more than 100 million m³ storage) and 1525 medium-scale reservoirs (more than 10 million m³ storage) with hydropower plants have been built in the end of 2016, and their total installed hydropower capacity is 200 GW, which accounts for over 60 % of the installed hydropower capacity (332 GW) in China.

Jina-Sha River located at the upstream of Yangtze River possesses the largest hydropower potential in the 13 large hydropower bases of China. The six mega cascade reservoirs have been constructed in the middle reach of Jin-Sha River (Figure 2 (a)) and are the pivotal hydropower bases for the China Southern Power Grid (http://eng.csg.cn/home/index.html). The mega reservoir is defined herein the reservoir with the total storage capacity greater than 100 million m^3 , the height of the dam more than 100 m, and the installed power capacity larger than 1000 MW. The climate in Jin-Sha River basin is the humid subtropical climate with the average annual rainfall of 736 mm, and the average annual runoff is 53 billion m³. The topography is high mountains with a large relief. Thanks to the humid climate and mountainous topography, this area has a high hydropower potential. The interannual variability of rainfall is high, with 65 % falls during flood season. The flood season generally

174 lasts from June to September. The mega cascade reservoirs which served as multiple purposes 175 not only can generate approximately 13.76 GW of hydropower (i.e. installed capacity) but 176 also can protect millions of downstream residents from flood hazards. These mega cascade 177 reservoirs have been managed to meet electricity demands of domestic and industrial sectors, 178 enable hydropower generation, and carry out flood control operation. The six cascade 179 reservoirs have total reservoir storage of 7.14 billion m³ and total watershed area of 250 180 thousand km² respectively. The characteristic parameters of cascade reservoirs are listed in 181 Table 2.

According to the Chinese Flood Control Act, reservoir water levels generally are not allowed to exceed the top of the buffer pool (see in Table 2) during flood season to provide adequate storage for flood prevention. During the impoundment operation period in the Jin-Sha River basin, the reservoir water level would be raised from the top of buffer pool on August 1st to the top of conservation pool (see in Table 2) by the end of October. If the reservoir water level is below the top of the conservation pool by the end of October, the water level rising would continue into November. From November to the end of May in the following year, the reservoir water level would generally be operated at the Zone I or II and it would be lowered gradually through control of the reservoir water release, which depends on the reservoir inflow (Zhou et al., 2014, 2015). As shown in Figure 2 (b), every reservoir authority has implemented the current operation rule curves (i.e. the standard operation policy (SOP)) to give guidance in hydropower generation (He et al., 2019). The guidance is described as follows.





1 2	Table 2 Characteristic parameters of cascade	reservoirs	s in the <mark>m</mark>	iddle Jin-	-Sha Riv	er reach	
2	Decemuin	Jin-Sha River Basin					
4	Reservoir	LY	AH	JAQ	LKK	LDL	GYY
5	Total storage capacity (Billion m ³)	0.81	0.89	0.91	0.56	1.72	2.25
6	Top of buffer pool (m)	1605	1493.3	1410	1289	1212	1128.8
./ o	Top of conservation pool (m)	1618	1504	1418	1298	1223	1134
o 9	Installed power capacity (GW)	2.40	2.00	2.40	1.80	2.16	3.00
.0	Minimum power capacity (GW)	0.41	0.29	0.50	0.33	0.43	0.57
1 2	Guaranteed power output (GW) of 6 cascade reservoirs	3.12					
.3	Flood season	June 1st to September 30th					
.4 .5 –	Non-flood season	October 1st to the next May 31st					

In Zone I (Power output < Guaranteed power output): the reservoir water release is equal to the reservoir inflow if the reservoir water level locates in the Zone I and the reservoir inflow is less than or equal to the water consumption corresponding to generating the guaranteed power output, otherwise the reservoir water release is equal to the water consumption corresponding to generating the guaranteed power output if the reservoir inflow is larger than the water consumption corresponding to generating the guaranteed power output. In Zone II (Guaranteed power output \leq Power output < Maximum power output): the reservoir water release is equal to the water consumption corresponding to generating the guaranteed power output if the reservoir water level locates in the Zone II.

In Zone III (Power output = Maximum power output): the reservoir would increase the water release to decrease the reservoir water level into Zone II in the next time step if the reservoir water level locates in the Zone III at the current time step.

Data used in this study consist of a total 65 742 (= 365 days (or 366 days) * 30 years * 6 reservoirs) reservoir inflow datasets collected in 30 hydrological years (June 1st-the next May 31st 1988-2018) at a temporal scale of day. The cascade reservoirs characteristics and inflow data are extracted from the Changjiang Water Resources Commission in China

(http://www.cjw.gov.cn/, in Chinese). Three hydrological scenarios (dry, normal, wet) are
designed to assess the impacts of different reservoir inflows on the hydropower output of
cascade reservoirs.

3. Methods

This paper proposes an improved KA to optimize the hydropower generation of the cascade reservoirs by introducing three auxiliary strategies. The improved KA can overcome the shortcomings of the standard KA encountered in the nonlinear and non-convex objective function as well as the high dimensional optimization operation of the cascade reservoirs. Figure 3 illustrates the architectures of the hydropower generation model (Figure 3 (a)), the standard KA (Figure 3 (b)) and the improved KA (Figure 3 (c)). The standard KA and GA served as the benchmark in this study. The methods used in this study are briefly introduced as follows.

232 3.1 Problems formulation of mega cascade reservoirs operation

The optimization operation of the cascade reservoirs is modelled for maximizing total hydropower generation equipped with the penalty function to avoid violations of the guaranteed (or firm) power output. The objective is to specify the optimal solution to maximize energy generation during the operation period in consideration of different operational and physical constraints. A sketch of the variables used to define the objective function and constraints is presented in Figure 3 (a). The objective function is defined to maximize hydropower generation:





265

$$HG = \text{maximize } \frac{1}{N} \sum_{t=1}^{T \cdot N} \left[\sum_{j=1}^{M} P_j(t) - \theta \cdot \left(\sum_{j=1}^{M} P_j(t) - P_g \right)^2 \right] \cdot \Delta t$$
(1a)

$$\theta = \begin{cases} 1, & if(\sum_{j=1}^{M} \mathbf{P}_j(t) < \mathbf{P}_g) \\ 0 & else \end{cases}$$
(1)

b)

$$P_j(t) = \eta_j(t) \cdot \rho \cdot g \cdot RT_j(t) \cdot H_j(t)$$
(1c)

$$\eta_j(t) = \varphi\left(\mathrm{RT}_j(t), \mathrm{H}_j(t)\right) \tag{1d}$$

where HG is the average annual hydropower generation of the cascade reservoirs. T is the number of time-steps in a year. N is the number of years. M is the number of reservoirs. Δt is the time-step. P_{g} is the guaranteed (or firm) power output of the cascade reservoirs. θ is the penalty factor, in which the value of θ is 1 on condition that the hydropower output of the cascade reservoirs is less than the guaranteed power output. $P_i(t)$ is the output power of the *j*th reservoir at the *t*th time. $RT_i(t)$ is the water release through the turbine of the *j*th reservoir at the *t*th time. $H_j(t)$ is the hydraulic head difference between the turbine intake and the last tank of the *j*th reservoir at the *t*th time. $\eta_j(t)$ is the dimensionless efficiency coefficient of the *j*th reservoir at the *t*th time and is a function $\varphi(\cdot, \cdot)$ of the water release and water head, in which the relation curve of efficiency coefficient $(\eta_j(t))$, water release $(RT_j(t))$ and hydraulic head $(H_i(t))$ can be found in the technical manual of the turbine developed by the manufacturers. ρ is the density of water. g is the gravity acceleration.

Reservoir operation should obey physical constraints containing the water balance equation, the hydraulic connection equation, the feasible boundary of the water release, the hydropower output and the reservoir water level. The mathematical formulations of these constraints are given as follows:

1266
$$V_j(t+1) = V_j(t) + \left[\frac{(t_j(t+1)+t_j(t))}{2} - \frac{(R_j(t+1)+R_j(t))}{2}\right] \cdot \Delta t$$
(2)267 $I_j(t+1) = R_{j-1}(t+1) + IF_j(t+1)$ (3)268 $R_j(t) = RT_j(t) + RS_j(t)$ (4a)10269 $R_j^{\min} \leq R_j(t) \leq R_j^{\max}$ (4b)11269 $R_j^{\min} \leq P_j(t) \leq P_j^{\max}$ (5)12270 $P_j^{\min} \leq P_j(t) \leq W_j^{\max}$ (6)13271 $W_j^{\min} \leq W_j(t) \leq W_j^{\max}$ (6)14272where $V_j(t), I_j(t)$ and $R_j(t)$ are the water volume, inflow and water release of the *j*-th reservoir14between the $(j-1)$ -th reservoir and the *j*-th reservoir at the $(t+1)$ -th time. $RS_j(t)$ is the water15276released through the spillway of the *j*-th reservoir, respectively. P_j^{\min} and P_j^{\max} are the16minimum and maximum water releases of the *j*-th reservoir, respectively. $W_j(t)$ is the17water level of the *j*-th reservoir at the *t*-th time. W_j^{\min} and W_j^{\max} are the17minimum and maximum power outputs of the *j*-th reservoir, respectively. $W_j(t)$ is the17water level of the *j*-th reservoir at the *t*-th time. W_j^{\min} and W_j^{\max} are the minimum and17maximum water levels of the *j*-th reservoir, respectively. The variables of the above equations17are non-negative.18In this study, the W_j^{\min} is equal to the top of the inactive pool in both the flood season and

282 non-flood season whereas the W_j^{max} is equal to the top of the buffer pool in the flood season

 $\tilde{60}$ 283 and the top of the conservation pool in the non-flood season respectively (Figure 2 (b) and

Table 2). Eqs. (2) and (3) are the water balance equation and hydraulic connection equation respectively. Eqs. (4), (5) and (6) show the constraints of water release, hydropower output and reservoir water level respectively. Furthermore, the water releases of the cascade reservoirs are selected as the decision variables of the optimization model.

3.2 Kidney algorithm (KA)

The KA proposed by Jaddi et al. (2017) has been found a quite successful state-of-the-art optimization algorithm suitable for tacking a wide variety of engineering applications, (e.g., Ekinci et al., 2018; Jaddi and Abdullah, 2018). As known, the kidneys play a vital role in filtering blood in the body. They filter blood to repel additional materials and surplus water from the body and blood present.

There are parts in the structure of kidneys which are called nephrons. Each kidney contains millions of nephrons. Every nephron is considered as a filtration unit. Kidneys manipulate following the four processes, i.e., filtration, reabsorption, secretion, and excretion. According to the analogy between the KA and the kidney biological system, Figure 3 (b) shows the flow diagram of the KA optimizing process. The implementation procedure is briefly described as follows:

Step 1: Initialization of feasible solutions and implementation of objective function evaluation. In a population of solutes, each solute within the blood present is taken as a candidate solution in the population of the algorithm. It is noted that each solute (or solution) is used to code the decision variables, i.e., the water release of the reservoir. For this study, real coded solutions are adopted, and then the objective function evaluation is implemented for each solution as well as ranking their values according to the descending sequence.

1 306

Step 2: Movement of Solutes (S). The movement operator is a process that the new solution (or solute) is produced through attempting to move a current solution toward the best solution based on the results of the objective function evaluation (Step 1), formulated as below:

$$S_{i+1} = S_i + rand(S_{best} - S_i)$$
⁽⁷⁾

where S_i is the solution in the population of the KA at the *i*-th epoch. S_{best} is the best solution at the current epoch. The value of *rand* is a random number between 0 and a given number (such as $(S_{best} - S_i)$).

Step 3: Filtration. The filtration operator is a process that the solutions in the population are filtered using a filtration rate through calculating a filtration function at each epoch. The filtrated solutes are moved to Filtrated Blood (FB) and the rest are transferred to Waste (W). In other words, if the objective function value of a solution is large than or equal to a filtration rate (fr), the solution will be transferred to a part of FB. Otherwise, it will be moved to a part of W. The filtration rate fr is formulated as below:

$$fr = \alpha \times \frac{\sum_{i=1}^{N_{p}} f(x_{i})}{N_{p}}$$
(8)

where *fr* is the rate of filtration. α is the filtration coefficient (constant number) in the range of (0, 1]. $f(x_i)$ is the objective function of solution *x* at *i*th epoch. N_p is the population size.

Step 4: Reabsorption. The reabsorption operator is a process that the solutions of W would be given a chance to turn into part of FB, owing to executing the movement operator (Eq. (7)) again, on condition that it meets the requirement of the filtration rate and then would be transferred to a member of FB.

Step 5: Secretion. The process of secretion is for the solutions, which have been moved to a part of FB after reabsorption. If one of the mentioned solutions has a lower quality as compared with the worst solution in FB, it would be secreted from the blood current and is classified as a part of W. Otherwise, it would be reserved in FB as well as the worst solution in FB is secreted and is turned into a part of W.

Step 6: Excretion. The excretion operator is a process that the solutions in W excreted if they cannot meet the requirement of the filtration rate for becoming a part of FB after implementation of reabsorption for them. Meanwhile, these solutions would be excreted on condition that they do not have the capability for turning into a part of FB after conducting movement operator twice. Under this circumstance, such a solution in W would be substituted by a random solution. Before moving toward the next epoch, the excretion is used to update the S_{best}, merges W and FB solutions, while recalculating the filtration rate. Terminate the computation process subject to the stopping criteria (early stopping or the maximal epoch E_{max}). In the case of the maximization hydropower generation problem, if the value of the objective function does not increase over 100 consecutive epochs, hydropower generation can no longer be enhanced, which triggers the computation to stop. If the epoch number is less than the maximum epoch "Emax", then repeat Steps 2-6. Otherwise, stop and output the optimization results.

The parameters of the KA consist of the maximum epoch (E_{max}) , the population size (N_p) and the filtration coefficient (α). The parameters of the KA could be obtained by using an intensive trial-and-error procedure for producing converged results.

3.3 Improved KA

Despite the KA has been demonstrated its success in coping with the reservoir optimization operation and other engineering applications, the KA, similar to other evolutionary intelligent algorithms, has the drawbacks of weak ability to identify the global optimal solution, especially in complex high-dimensional cascade reservoirs optimization operation with a nonconvex function, a huge number of constraints and decision variables. In other words, the KA would demand auxiliary strategies to increase the performance and flexibility to cope with complex and real-world optimization problems. Therefore, to improve the ability to obtain the global optimal solution, three auxiliary strategies, i.e., the exploration and exploitation strategy for stimulating global optimization ability, the adaptive strategy for adjusting filtration coefficient and the elitist strategy for storing best solution, are fused into the standard KA in this study. The three strategies were briefly described as below.

3.3.1 Exploration and exploitation strategy for stimulating global optimization ability

It is worth noting that the Eq. (7) could not offer a high diversity of solutions for promoting the global exploration capability and local exploitation ability, because the solutions only varied based on the current solution (S_i) and the best solution (S_{best}). Bearing this in mind as a motivation, the exploration and exploitation strategy is accordingly applied to stimulate the movement of solutions (or maneuver of solutions). One makes use of the current solution and a weighted difference between the best solution and random solutions to boost the global exploration ability. Another makes use of the best solution and a weighted difference between the current solution and random solutions to facilitate the local exploitation capability. The proposed exploration and exploitation strategy is formulated as below.

Global exploration strategy

$$S_{global} = S_{best} + \beta \cdot rand(|S_i - S_{FB}|) + (1 - \beta) \cdot rand(|S_i - S_W|)$$
(9a)

372 *Local exploitation strategy*

$$S_{local} = S_i + \beta \cdot rand(S_{best} - S_{FB}) + (1 - \beta) \cdot rand(S_{best} - S_W)$$
(9b)

374 *Combination of exploration and exploitation strategy*

$$S_{i+1} = \gamma \cdot \max\left(S_{global}, S_{local}\right) + (1 - \gamma) \cdot \min\left(S_{global}, S_{local}\right)$$
(9c)

where S_{global} and S_{local} are the solutions raised by the exploration and exploitation strategy, 376 respectively. β and γ are the random numbers in the range of (0, 1). S_{FB} and S_{W} are the 377 random solutions in the part of FB and W, respectively, in which $S_{FB} \neq S_i \neq S_W$. In 378 comparison to Eq. (7), the Eq. 9 (a) can be useful for global exploration by taking full 379 advantage of the information difference between the best solution and the random solutions of 380 FB & W, whilst the Eq. 9 (b) can be beneficial to local exploitation by making full use of the 381 information difference between the current solution and the random solutions of FB & W. 382 That is to say, the combination of exploration and exploitation strategy (Eq. 9 (c)) not only 383 384 can be applied to direct at the avoidance of low diversity and trapping into a local optimum but also can make a suitable tradeoff between the exploration and exploitation within the 385 search domain for achieving the global optimum. 386

387 *3.3.2 Adaptive strategy for adjusting filtration coefficient*

It is also worth noting that the filtration coefficient (α) of the standard KA in Eq. (8) is a constant value in the range of (0, 1], which is given in advance. In general, the constant parameter values have a substantial impact on the quality of the solutions and the robustness of evolutionary algorithm (Srinivas and Patnaik, 1994; Molina et al., 2018). Additionally, the

392

selection of appropriate parameter values is usually resolved by the trial-and-error procedure and demands the developers and users' prior knowledge, in which the process is timeconsuming due to the sensitivity analysis of adjusting algorithm parameters. To conquer such technical bottleneck, the adaptive strategy for adjusting algorithm parameter values were adopted by a variety of researches and was widely used to enhance the quality of the solutions and the robustness of evolutionary algorithms (e.g., Zhang et al., 2007; Zhou et al., 2017). Owing to its reliability and wide practicality, the adaptive strategy for adjusting filtration coefficient is also integrated into the KA in this study and is formulated as below:

$$\alpha = \begin{cases} \varepsilon_1 \cdot \left[\left(f(\mathbf{S}_{\text{best}}) - f(\mathbf{S}_{\text{FB}}) \right) / \left(f(\mathbf{S}_{\text{best}}) - f_{\text{avg}} \right) \right], & if(f(\mathbf{S}_{\text{FB}}) \ge f_{\text{avg}}) \\ \varepsilon_2, & otherwise \end{cases}$$
(10a)

$$f_{\text{avg}} = \frac{\sum_{i=1}^{N_{\text{p}}} f(x_i)}{N_{\text{p}}}$$
(10b)

where ε_1 and ε_2 are the random numbers in the range of (0, 1]. f_{avg} is the average value of the objective function in the KA. $f(S_{FB})$ and $f(S_{best})$ are the objective function values of the random solution in the FB and the best solution, respectively.

3.3.3 Elitist strategy for storing best solution

The concept of elitism proposed by Goldberg (1989) intends to avoid the algorithm getting stuck in local optimal solutions, and the elitist strategy has been widely adopted for improving the performance of the evolutionary algorithms, for instance, GA (Goldberg, 1989; Wardlaw and Sharif, 1999), NSGA-II (Deb et al., 2002), PSO (Bai et al., 2017) and BA (Bora et al., 2012). The Eq. (9) could provide the KA with a high diversity of solutions, whereas both the Eqs. (7) and (9) could not guarantee that the good solutions would not be discarded even if they have been found before reaching the maximum epoch. Therefore, in this study, if the

solution created in the previous epoch (S_{i-1}) is not better than the current solution (S_i) , the elitist strategy will be used with a certain probability. Inspired by the concept of elitism, the proposed strategy employs the best solution (S_{best}) and a difference between the current and random solutions $(S_{global}, shown in Eq. 9 (a))$ for lifting the performance of the KA to prevent the loss of good solutions once they are found, which is formulated as below:

In the case of maximization problem:

$$S_{i} = \begin{cases} S_{i}, & if(f(S_{i}) \ge f(S_{i-1})) \\ S_{global}, & else \ if(f(S_{i}) < f(S_{i-1}) \ and \ \beta < 0.5) \\ S_{i-1}, & otherwise \end{cases}$$
(11a)

In the case of minimization problem:

$$S_{i} = \begin{cases} S_{i}, & if(f(S_{i}) \leq f(S_{i-1})) \\ S_{global}, & else \ if(f(S_{i}) > f(S_{i-1}) \ and \ \beta < 0.5) \\ S_{i-1}, & otherwise \end{cases}$$
(11b)

where $f(S_i)$ is the value of objective function of the solution at the *i*th epoch. The Eq. (10) equipped with the elitist strategy can be used to avoid the loss of good solutions. That is to say, the good solutions would be stored when they have been found before meeting the requirement of the maximum epoch.

The following section describes how to fuse the three auxiliary strategies into the standard KA for optimizing the cascade reservoirs operation. Figure 3 (c) shows the flow diagram of the improved KA optimizing process. The implementation procedure is described as follows.

430 Step 1: Initialization of feasible solutions and implementation of objective function
431 evaluation. Because none of the auxiliary strategies has been implemented for this step, this
432 process could refer to the Step 1 in the standard KA.

Step 2: Movement of Solutions (or maneuver of Solutions) (S) using the exploration and exploitation strategy. According to the rankings of the objective function (Step 1), the improved movement operator (Eq. (9)) would be conducted to promote the movement of S.

Step 3: Filtration using adaptive strategy. The improved filtration operator (Eq. (10)) will be implemented for dividing the S into the two parts of the FB and W.

Step 4: Reabsorption. The reabsorption operator would be executed to render an opportunity for the solutions of W transferring into a part of FB if it satisfies the condition of the filtration rate. And then the improved movement operator (Eq. (9)) would be run once again in this procedure. That is to say, the course of the reabsorption can also be enhanced due to the improved movement of S.

Step 5: Secretion. This process can refer to the Step 5 in the standard KA.

Step 6: Excretion and implementation of the elitist strategy. The excretion operator would also be carried out if the solutions in W cannot meet the requirement of the filtration rate for becoming a part of FB. In addition, the elitist strategy (Eq. (11)) would be conducted to store the best solution. Terminate the computation process subject to the stopping criteria (early stopping or the maximal epoch E_{max}). For the maximization hydropower generation problem, when the value of the objective function does not increase over 100 consecutive epochs, hydropower generation can no longer be improved, which induces the computation to stop. When the maximum epoch " E_{max} " is reached, the computation process stops and outputs the optimization results. Otherwise, update the epoch and repeat Steps 2-6.

As compared with the standard KA, the merits of the improved KA consist of: firstly, in Step 2, the combination of exploration and exploitation strategy (Eq. 9 (c)) not only can conquer the bottlenecks of low diversity and trapping into a local optimum but also can make an adequate balance between the exploration and exploitation for searching the global optimum; secondly, in Step 3, the adaptive strategy is utilized for adjusting the filtration coefficient parameter to overcome the time-consuming encountered in the trial-and-error procedure (or sensitivity analysis) of selecting appropriate parameter values; lastly, in Step 6, the elitist strategy is used to avoid the loss of good solutions before reaching up to the maximum epoch.

4. Results and discussion

The results and findings are presented and discussed in details in the order of three parts: the sensitivity analysis of evolutionary algorithm parameters (GA served as the benchmark) as well as the comparison between the KA and the improved KA (KA served as the benchmark), and the summarization, shown as follows.

4.1 Sensitivity analysis of GA and KA parameters

In this section, special attention is paid to the extension of the KA to the optimization of mega cascade reservoirs at a time scale of day. The GA serves as a benchmark. And the parameters of the GA consist of the population size (N_p) , the maximum epoch (E_{max}) , the crossover probability (P_c) and the mutation probability (P_m). The sensitivity analysis of evolutionary algorithm parameters is conducted for the optimization operation of the six cascade reservoirs in the Jin-Sha River basin (Figure 2). Each evolutionary algorithm is driven by a total of 65742 (= 365 days (or 366 days) * 30 years * 6 reservoirs) datasets, which means we have 65742 decision variables and 262968 constraints (= 4 equations * 65742 decision variables).

For the GA, various researches (e.g., Wardlaw and Sharif, 1999; Deb et al., 2002) have suggested that for complex cascade reservoirs system, a larger value of N_p is required to maintain the diversity in the population; a larger value of E_{max} is required to converge to a state at which there are no changes in the objective function value over 100 generation; good performance can be achieved using a high value of P_c and low value of P_m. For the KA, to obtain good performance, the parameter of the filtration coefficient (α) is additionally advised to use a medium-low value (Ehteram et al., 2018a, b). Therefore, on condition that both the KA and GA used the same population size ($N_p = 500$) and maximum epoch ($E_{max} = 1000$), we concentrate on the following sensitivity analysis: for the GA, the most appropriate Pc and Pm would appear to be in the range of 0.75 up to 0.95 and 0.05 up to 0.25 at an increasing step of 0.05, respectively; for the KA, the most appropriate α would appear to be in the range of 0.25 up to 0.55 at an increasing step of 0.05.

The results of the sensitivity analysis of the GA and KA parameters are shown in Figure 4. Figure 4 (a) indicates a distinct peak in performance at $P_c = 0.85$ as well as progressive deterioration in performance as the value of P_c increases beyond this, whilst there is a distinct peak in performance at $P_m = 0.10$ as well as progressive deterioration in performance as the value of P_m increased beyond this. That is to say, the most appropriate values of P_c and P_m are 0.85 and 0.10, respectively.

The Figure 4 (b) reveals that the best result (= 0.977) in the KA is achieved with the value of α (= 0.35) using the population size (N_p = 500) and maximum epoch (E_{max} = 1000) whereas there is a progressive deterioration in performance as the value of α increased beyond this. It needs to take about 2.1 hours and 1.3 hours computation time (mean of 10 runs of each



Figure 4 Sensitive to optimization algorithm parameters and optimization progress using the population size ($N_p = 500$) and maximum epoch ($E_{max} = 1000$). **a.** Sensitive to the crossover (P_c) and mutation (P_m) probability of GA. **b.** Sensitive to the filtration coefficient (α) of KA. **c.** Optimization progress in GA using the most appropriate parameters ($P_c = 0.85$, $P_m = 0.10$, N_p = 500 & $E_{max} = 1000$) and KA using the most appropriate parameters ($\alpha = 0.35$, $N_p = 500$ & $E_{max} = 1000$). The computation result is the average result of 10 runs of each algorithm and the objective function value is normalized between 0 up to 1.

)7	evolutionary algorithm) for the implementation of the GA and KA to optimize the operation
)8	of the six cascade reservoirs, conducted by a DELL computer (Intel® CoreTM i5, 7th
)9	Generation CPU @ 2.50 GHz, RAM 8 GB and 1 TB Hard Disk). That is to say, in each trial-
LO	and-error computation process, it spends approximately 2.1 hours for the GA to find the
11	appropriate parameters of P_c (or P_m) whereas it spends 1.3 hours for the KA to search the
12	appropriate parameter of α . The most appropriate parameters of the GA are set as: $N_p = 500$;
L3	$E_{max} = 1000$; $P_c = 0.85$; and $P_m = 0.10$. The most appropriate parameters of the KA are set as:
L4	$N_p = 500$; $E_{max} = 1000$; and $\alpha = 0.35$. Table 3 summarizes the computation results of the
L5	evolutionary algorithms (GA & KA) in terms of 10 runs of the GA and KA using the most
L6	appropriate parameters. Firstly, from the standpoint of the final objective function value
L7	(normalization), the KA produces much higher final objective function values than the GA in
L8	terms of the best (0.981), average (0.977) and worst (0.974) final objective function values. At
19	the same time, the standard deviation value of the final objective function in KA is equal to
20	0.0027, which is noticeably smaller than that (0.0042) of the GA. That means the robustness
21	of the KA is stronger than that of the GA. Secondly, from the standpoint of the convergence
22	speed, the number (mean = 406) of epoch attained the convergence result is significantly less
23	than that (mean = 561) of the GA. Such results demonstrate fewer epochs for the KA are
24	required to search out the optimal solution (shown in Figure 4 (c)). Thirdly, from the
25	standpoint of the hydropower generation, the global optimal solution obtained from the KA
26	can largely improve hydropower generation by 2.9 billion kW·h/year and 1.5 billion
27	kW·h/year accordingly. In addition, the improved KA can achieve 64.6 kW·h/year
28	hydropower generation (SOP, 61.7 billion kW·h/year and GA, 63.1 billion kW·h/year). The

improvement rates reach 4.7 % and 2.4 %, respectively. The reasons for the KA's superior performance than that of the GA consist of: firstly, the filtration operator of the KA provides the algorithm with good exploitation and fast convergence in comparison to the selection operator of the GA; secondly, the movement and reabsorption operator of the KA gives the algorithm a good diversity of solution and thus superior exploration as compared with the crossover and mutation operators of the GA.

Table 3 Computation results of the evolutionary algorithms (GA & KA)

	normalization			
Number of runs	final objective fu	inction value		
	KA	GA		
1	0.975	0.963		
2	0.974	0.961		
3	0.981	0.968		
4	0.979	0.966		
5	0.974	0.971		
6	0.977	0.973		
7	0.981	0.971		
8	0.976	0.968		
9	0.980	0.973		
10	0.977	0.964		
Mean	0.977	0.968		
Best	0.981	0.973		
Worst	0.974	0.961		
Standard deviation	0.0027	0.0042		
Mean of time cost (Hours)	1.3	2.1		
Average annual hydropower generationa (Billion kW·h)	64.6	63.1		
Average annual hydropower generation (Billion kW [·] h) using the SOP ^b	61.7	7		
Most appropriate parameters	$N_{p} = 500$ $E_{max} = 1000$ $\alpha = 0.35$	$N_{p} = 500$ $E_{max} = 1000$ $P_{c} = 0.85$ $P_{m} = 0.10$		

Note: The daily data from June 1st 1988 up to May 31st 2018 (30 hydrological years) are used in this study.

^a The hydropower generation is the average annual hydropower generation during 1988 and 2018 and is the average result of 10 runs of each algorithm.

^b SOP is the Standard Operating Policy using operation rule curves.

4.2 Comparison between KA and improved KA

In the case of six cascade reservoirs operation, the computation results (average results of ten

runs) of the four schemes concerning the KA and improved KA are reported in Table 4. It is

noted that: firstly, the difference between KA0 and KA1 (using one auxiliary strategy) is that the latter uses the exploration and exploitation strategy whereas the former does not; secondly, the difference between KA1 and KA2 (using two auxiliary strategies) is that the latter adopts the adaptive strategy for adjusting filtration coefficient whereas the former does not; and lastly, the difference between KA2 and KA3 (using three auxiliary strategies) is that the latter employs the elitist strategy for storing best solution whereas the former does not.

Table 4 Computation results of the four schemes concerning the standard KA and improved KΔ

5	NА					
		Scheme	KA0	KA1	KA2	KA3
		N_p	500	500	500	500
	Parameters	E_{max}	1000	1000	1000	1000
	Parameters Parameters Auxiliary strategy Number of obj Me Average annual hy Average annual hydrop Note: The computation res 1988 up to May 31 st 2018 KA0: the optimization alg	α	0.35	0.35	/	/
		Exploration and exploitation strategy	/	Yes	Yes	Yes
	Auxiliary strategy	Adaptive strategy	/	/	Yes	Yes
		Elitist strategy		/	/	Yes
	Number of obj	406	451	539	487	
	Me	1.3	1.5	1.8	1.6	
	Average annual h	ydropower generation ^a (Billion kW·h)	64.6	65.1	65.6	66.5
	Average annual hydrop	ower generation (Billion kW [.] h) using SOP ^b	61.7			
4	Note: The computation re	sult is the average result of 10 runs of each algo	orithm and	l daily da	ta from J	une 1 st
5	1988 up to May 31 st 2018					
6	KA0: the optimization alg	orithm is the standard KA.				
7 KA1: the optimization algorithm is the improved KA with one auxiliary strategy.						

41 558 KA2: the optimization algorithm is the improved KA with two auxiliary strategies.

KA3: the optimization algorithm is the improved KA with three auxiliary strategies.

^a Average annual hydropower generation is the mean of annual hydropower generated during 1988 and 2018.

^b SOP is the Standard Operating Policy using operation rule curves.

 4.2.1 Hydropower generation

The results in Table 4 indicate that: firstly, in comparison to the SOP (61.7 billion kW·h/year), the improved KA1 can increase the hydropower generation 3.39 billion kW·h/year (5.5 % improvement), owing to the exploration and exploitation strategy; secondly, the improved

KA2 can lift the hydropower generation 3.89 billion kW·h/year (6.3 % improvement), in the combination of the exploration and exploitation strategy as well as the adaptive strategy; lastly, the improved KA3 can enhance the hydropower generation 4.81 billion kW·h/year (7.8 % improvement) due to integration of the three auxiliary strategies. As compared with the KA0, the improved KA3 can promote the hydropower generation of 1.91 billion kW·h/year (3.0 % improvement). That is to say, the use of the three auxiliary strategies, the improved KA can dramatically enhance the hydropower generation in virtue of finding the global optimal solution. The average time cost (2.2 hours) of the improved KA is higher than the KA (1.3 hours), whereas the improved KA can save a lot of time searching appropriate algorithm parameter due to using the adaptive strategy for adjusting the filtration coefficient. That is to say, the improved KA not only can increase the hydropower generation but also can conquer the time-consuming encountered in the trial-and-error procedure (or sensitivity analysis) of selecting appropriate parameter values, in comparison to the standard KA.

To show the merits of the improved KA, an assessment is conducted on the results obtained from the convergence process of the four schemes (KA0-KA3) for optimization operation of the six cascade reservoirs (Figure 5). The comparison between KA0 and KA1 (with one auxiliary strategy) shows that the final objective function value (0.985) of the improved KA1 is considerably larger than that (0.977) of the KA0. The combination of exploration and exploitation strategy (Eq. 9 (c)) not only can boost solution diversity and escape the trap of a local optimum but also can increase the objective function (i.e. hydropower generation). Moreover, the objective function values of the KA1 show more fluctuation than those of KA0, which implies the KA1 would easily trigger optimization



process instability problem due to the utilization of the exploration and exploitation strategy.

б

The results indicate that the KA required more auxiliary strategies to handle its instability problem. The comparison between KA1 and KA2 (with two auxiliary strategies) shows that the objective function values of the KA2 fluctuated less and are moderately larger than those of the KA1, which demonstrates the KA2 can overcome the instability in virtue of the adaptive strategy for adjusting algorithm parameter values. The reason is that the adaptive strategy can dynamically adjust the parameter values in response to the higher solution diversity produced by the exploration and exploitation strategy. The comparison between KA2 and KA3 (with three auxiliary strategies) shows that the final objective function value (0.996) of the KA3 is considerably larger than that (0.987) of the KA2. The KA3 can converge faster and is more robust as shown in Figure 5 and Table 3. The faster convergence and better robustness is the result of the good exploration and exploitation provided by the integration of the three auxiliary strategies.

The comparative results demonstrate that the improved KA not only best optimizes hydropower generation with fast convergence as well as the most stable objective function curve, but also can effectively conquer the shortcomings of trapping into local optimums, instability and loss of good solutions. This is due to the utilization of the exploration and exploitation strategy, the adaptive strategy as well as the elitist strategy.

4.2.2 Reliability, vulnerability and resilience of hydropower output

A coherent set of evaluation criteria is used to distil the merits of the improved KA to quantitatively assess the impacts and contributions of the KAs on the hydropower generation in different periods (year-round, flood season, non-flood season) and hydrological representative years (dry, normal, wet). The criteria are designed for assessing the reliability,

vulnerability and resilience of hydropower output (Hashimoto et al., 1982; Zhou et al., 2017).
Their formulations are given as follows.

Reliability of hydropower output: The reliability can be described by the probability that a hydropower energy system remains in a satisfactory state.

Reliability =
$$\frac{n - \sum_{t=1}^{n} NT(t)}{n}$$
 (12a)

$$NT(t) = \begin{cases} 1 & if(\sum_{i=1}^{M} P_i(t) < P_g) \\ 0 & else \end{cases}$$
(12b)

where NT(t) is the number of time that total hydropower output is less than the guaranteed power output of the cascade reservoirs at the *t*-th time. n (=N·T) is the total number of time steps in the operation period.

Vulnerability of hydropower output: The vulnerability represents the incompetence of a
hydropower energy system to resist the effect of a hostile environment. It denotes the
maximum ratio of hydropower output deficiency to installed power capacity if once occurs,
shown as follows.

$$Vulnerability = \max_{1 \le t \le n} VU(t)$$
(13a)

$$VU(t) = \begin{cases} \frac{P_g - \sum_{i=1}^{M} P_i(t)}{P_g} & if(\sum_{i=1}^{M} P_i(t) < P_g) \\ 0 & else \end{cases}$$
(13b)

where VU(t) is the vulnerability of hydropower output at the *t*-th time.

Resilience of hydropower output: The resilience describes how quickly a hydropower system is likely to recover once hydropower output deficiency has occurred, shown as follows.

$$\text{Resilience} = \begin{cases} 1 & if(\text{Reliability} = 1) \\ \frac{\sum_{t=1}^{n-1} \text{RE}(t)}{\sum_{t=1}^{n} \text{NT}(t)} & else \end{cases}$$
(14a)

$$\operatorname{RE}(t) = \begin{cases} 1 & if(\left(\sum_{i=1}^{M} \mathsf{P}_{i}(t) < \mathsf{P}_{g}\right) \text{ and } \left(\sum_{i=1}^{M} \mathsf{P}_{i}(t+1) \ge \mathsf{P}_{g}\right) \\ 0 & else \end{cases}$$
(14b)

where RE(t) is the number of times that the hydropower energy system is likely to recover from hydropower output deficiency at the *t*-th time. The higher index value of reliability and resilience, as well as the lower index value of vulnerability, indicate better model performance. The index values of reliability, vulnerability and resilience in different scenarios are depicted in Table 5. From the standpoint of different periods (year-round, flood season & nonflood season), the results indicate that the improved KA can rapidly increase the index values of reliability (from 0.95 to 0.98) and resilience (from 0.87 to 0.93), and decrease the index value of vulnerability (from 0.11 to 0.07) in the case of year-round, as compared with the standard KA. Additionally, shown by the comparison with the SOP, the improved KA not only can raise the reliability and resilience with the improvement rates of 8.0 % and 14.8 % respectively but also can dramatically reduce the vulnerability by 46.7 % in the case of the non-flood season. Such substantial improvement is mainly owing to the good performance of the improved KA whilst the objective function of maximization hydropower generation closely linked with the guaranteed power output (Eq. (1)) also contributed to such improvement. Some interesting characteristics in different periods can be found in Table 5. For example, in all cases (SOP, KA & improved KA), both the index values of reliability and resilience in flood season are equal to one, while the index value of vulnerability in flood season are equal to zero. In other words, in flood season, the hydropower output of the six cascade reservoirs is always larger than or equal to the guaranteed power output (3.12 GW, in Table 2). Both the index values of resilience and vulnerability in the non-flood season are equal to both the index values of resilience and vulnerability in year-round. The index value

of reliability in year-round is always larger than the index value of reliability in non-flood season. The reason is the ratio of runoff in flood season to annual runoff ranges between 60 % and 70 % in this study area so that the hydropower output deficit always occurred in the nonflood season whereas both the reliability and resilience of hydropower output in flood season would reach up to 100%. In flood season, the potential of hydropower generation is driven by lessening the gap between hydropower output and installed (maximum) power capacity. However, in non-flood season, the potential of hydropower generation is driven by lessening the gap between hydropower output and guaranteed power output to improve the hydropower

generation.

Table 5 Com	putation r	esults of	f the K.	A and in	nproved	KA in	the d	lifferent	scenarios

Sahama	Indicators		Different periods				
Scheme	mulcators	Year-round ^a	Flood season ^b	Non-flood season ^c			
	Reliability	0.92	1	0.88			
SOP	Vulnerability	0.21	0	0.21			
	Resilience	0.81	1	0.81			
	Reliability	$0.95 (3.3 \%^{d})$	1	0.92 (4.5 %)			
KA	Vulnerability	0.11 (26.7 %)	0	0.11 (26.7 %)			
	Resilience	0.87 (7.4 %)	1	0.87 (7.4 %)			
T 1 TZ A	Reliability	0.98 (6.5 %)	1	0.95 (8.0 %)			
Improved KA (i.e. $KA3$)	Vulnerability	0.07 (46.7 %)	0	0.07 (46.7 %)			
(1.0. 1413)	Resilience	0.93 (14.8 %)	1	0.93 (14.8 %)			
Sahama	Indiantons	Hydrological representative years					
Scheme	mulcators	Dry ^e	Normal ^f	Wet ^g			
	Reliability	0.92	0.95	1			
SOP	Vulnerability	0.34	0.22	0			
	Resilience	0.72	0.80	1			
	Reliability	0.95 (3.2 %)	0.97 (2.1 %)	1			
KA	Vulnerability	0.20 (41.2 %)	0.14 (36.4 %)	0			
	Resilience	0.79 (9.7 %)	0.86 (7.5 %)	1			
	Reliability	0.97 (5.4 %)	0.99 (4.2 %)	1			
Improved KA $(i \in KA3)$	Vulnerability	0.15 (55.9 %)	0.11 (50.0 %)	0			
(I.C. KA3)	Resilience	0.85 (18.1 %)	0.92 (15.0 %)	1			

Note: The computation result is the average result of 10 runs of each algorithm and the daily data from June 1st 1988 up to May 31st 2018 (30 hydrological years) are used in this study.

^a Year-round is the hydrological year, starting from June 1st to the next May 31st in this study area.

^b Flood season: starting from June 1st to September 30.

^c Non-flood season: starting from October 1st to the next May 31st.
^d Improvement rate = <u>Indicator(Evolutionary Algorithm)-Indicator(SOP)</u> × 100%
^e Occurrence frequency of the dry year (2008) is 95% during 1988 and 2018.
^f Occurrence frequency of the normal year (2003) is 50% during 1988 and 2018.
^g Occurrence frequency of the wet year (2012) is 10% during 1988 and 2018.

Table 5 also shows the sensitivity of reliability, vulnerability and resilience of hydropower output in response to the hydrological representative years (dry, normal, wet). The results indicate that: as compared with the SOP, the improved KA can noticeably increase the reliability and resilience as well as decrease the vulnerability in dry and normal years. The improvement rates of reliability (from 0.92 to 0.97, 5.4 % improvement), vulnerability (from 0.34 to 0.15, 55.9 % improvement) and resilience (from 0.72 to 0.85, 18.1 % improvement) are higher especially in dry year. In all cases (SOP, KA & improved KA), both the index values of reliability and resilience in the wet year (2012, 10% occurrence frequency during 1988 and 2018) are equal to 1, while the index value of vulnerability in the wet year is equal to 0. In other words, in the wet year, the hydropower output of six cascade reservoirs is always larger than or equal to the guaranteed power output (3.12 GW, in Table 2).

4.2.3 Reservoir operation curves

Take the first reservoir (LY, Figure 2) and the last reservoir (GYY, Figure 2) of cascade reservoirs for example, Figure 6 presents the differences in the reservoir water level, water release and hydropower output trajectories generated by the KA and improved KA in the scenario of the dry year (2008, 95% occurrence frequency during 1988 and 2018). It can be further seen from Figures 6 (a) that: for flood season all the three trajectories are satisfied with the requirements of their constraints whilst for non-flood season sometimes dissatisfied the hydropower constraint in which the total hydropower output of two cascade reservoirs is less than the guaranteed (minimum) power output. Despite the violation of the constraint has
occurred in both the KA and improved KA, the times (1 time in both two cascade reservoirs)
generated by the improved KA is less than the times (2 times in both two cascade reservoirs)
generated by the KA in the scenario of the dry year (marked in red circle).

For flood season, the differences in the three trajectories generated by the standard KA and improved KA are small. The reservoir water level and hydropower output generated by the improved KA are slightly higher than that of the KA, whilst the water releases generated by the improved KA are briefly smaller than those of the KA in both two cascade reservoirs. For non-flood season the differences in the three trajectories generated by the KA and improved KA are considerable, in which the reservoir water levels and hydropower outputs generated by the improved KA are sharply higher than that of the KA whilst the water releases generated by the improved KA are sharply smaller than that of the KA in both 2 cascade reservoirs. In other words, for flood season the differences in the three trajectories generated by the KA and improved KA are small whereas for the non-flood season the differences in the three trajectories generated by the KA and improved KA are noticeable.

More interesting characteristic of the optimal hydropower output can be found in this study, for example, most of hydropower output trajectories generated by the improved KA are larger than or equal to those of the KA whereas small minority of hydropower output generated by the improved KA is less than that of the KA in both 2 cascade reservoirs (marked in red rectangle). Therefore, the improved KA can rapidly increase the reliability of hydropower output in comparison to the KA. The difference in the hydropower output trajectories can demonstrate the performance of the improved KA is noticeably superior to the



Figure 6 Comparison of optimal trajectories generated by the KA and improved KA with respective to the LY and GYY reservoirs in a dry year (2008) as well as theoretical relationship curves between power output, hydraulic head, and water consumption of a hydro unit (i.e., unit performance curves). a. Comparison of optimal trajectories. b. Hydro unit performance curves.

performance of the KA in the interests of hydropower generation maximization, whereas the differences in the reservoir water level or water release trajectories generated by the KA and improved KA does not necessarily indicate the superiority of one approach over the other. The reason is that: according to Eq. (1(c)), the value of hydropower output not only is not only dependent on the values of the water release (RT_j(t)) and hydraulic head (H_j(t)), but is also dependent on the value of efficiency coefficient ($\eta_i(t)$). The function $\varphi(RT_j(t), H_j(t))$ (Eq.

1(d)) is not monotonically increasing with the values of the water release (or water consumption) $(\text{RT}_j(t))$ and hydraulic head $(\text{H}_j(t))$ (Figure 6 (b)).

In summary, these comparative results demonstrate that the improved KA with three auxiliary strategies not only can produce the largest objective function values and the most stable objective function curve but also can effectively increase hydropower generation of mega cascade reservoirs. Such achievement made by the KA3 could be owing to that the exploration and exploitation strategy improved the hydropower generation from the perspective of tacking the technical bottleneck of trapping into local optimums, the adaptive strategy improved the hydropower generation from the perspective of conquering the instability of optimization process, while the elitist strategy improved the hydropower generation from the perspective of overcoming the loss of good solutions. Additionally, the indexes of reliability, vulnerability and resilience are used to assess the KAs for different periods (year-round, flood season, non-flood season) and different hydrological representative years (dry, normal, wet) comprehensively. Compared with the SOP, the improved KA can increase the index values of reliability and resilience, and decrease the index value of vulnerability in different periods (year-round & non-flood season) and different hydrological

representative years (dry year & normal year). The improvement rates of reliability, resilience and vulnerability are higher especially in non-flood season and dry year. The reason is that the probability of hydropower output deficit occurrence in the non-flood season and the dry year is higher than the probability of deficit occurrence in the flood season, and the normal & wet years. From the standpoint of hydropower benefits and CO₂ emission reduction, according to the hydropower price in China (45.3 USD/MW[•]h) and CO₂ emission reduction for hydropower production (0.785 kg CO₂ equivalent/kW[·]h) (Zhou et al., 2018a,b), in comparison to the SOP, the improved KA can dramatically stimulate the hydropower benefits 217.44 million USD/year (= 4.8 billion kW·h * 45.3 USD/MW·h) as well as reduce the CO₂ emission 3.77 billion kg/year (= 4.8 billion kW·h * 0.785 kg CO₂ equivalent/kW·h), respectively. To support the official mission - to fulfil the pledge of carbon emission reduction and non-fossil energy expansion to 20% in China by 2030 or earlier, this study indicates the niche and potential of the hydroelectricity as a guideline for the cleaner production.

In comparison to the dynamic programming methods, for instance, discrete differential dynamic programming, progressive optimality algorithm and dynamic programming successive approximation, the major advantage of the KA approach is that it does not demand the initial trial water release policy (Ehteram et al., 2018a, b), which can motivate the robustness of the algorithm and the stochasticity of solutions. As compared with the GA and KA, the main merit of the improved KA is the capability to find the global optimum with faster convergence speed. The reasons are as follows: firstly, the filtration operator provides the required exploitation while the reabsorption operator gives the necessary exploration for the evolutionary algorithm; the combination of exploration and exploitation strategy not only can conquer the bottlenecks of low diversity and trapping into a local optimum, but also can make an adequate balance between the exploration and exploitation for searching the global optimum; secondly, the adaptive strategy can automatically adjust the filtration coefficient parameter to overcome the time-consuming encountered in the trial-and-error procedure of selecting appropriate parameter values; lastly, the elitist strategy can avoid the loss of good solutions before reaching up to the maximum epoch.

5. Conclusion

In China, the developing hydroelectricity can provide a reliable and practical pathway in the transition to the low carbon and cleaner production for sustainable development. The optimization operation of mega cascade reservoirs can better produce hydropower outputs. However, the difficulty encountered in this process raises quickly since the number of cascade reservoirs, decision variables and constraints grow, in which the optimization process is easy to give rise to time-consuming and loss of good solutions as well as a trap into local optimum. In this study, we explored the KA with three auxiliary strategies for stimulating the hydropower output of cascade reservoirs. The standard KA, GA and SOP were selected as the benchmark for the comparison analysis. The improved KA was introduced to optimize the hydropower generation of six mega cascade reservoirs located at middle reach of Jin-Sha River in China. The mathematical model is driven by a huge number of inputs (i.e., 65742 inflow measurements and decision variables) and constraints (i.e., 262968 conditions).

Here we show that there is a great potential for application of the KAs to complex mega

cascade reservoir operation. As compared with the SOP, the KA and improved KA can increase the hydropower generation 2.9 billion kW·h/year (4.7 % improvement) and 4.8 billion kW·h/year (7.8 % improvement) while boost the hydropower benefits 131.37 million USD/year and 217.44 million USD/year as well as decrease the CO₂ emission 2.28 billion kg/year and 3.77 billion kg/year, respectively. Additionally, the improved KA can increase the index values of reliability and resilience as well as decrease the index value of vulnerability. The limitation of the KAs is that if a multipurpose reservoir operation is taken into consideration, it demands to reconstruct the optimization mechanism from a single objective into multi-objective optimization to find the Pareto-optimal solutions. Consequently, followup studies will fuse the non-dominated sorting strategy and/or dynamically dimensioned search into a Multi-objective Kidney Algorithm for optimizing the multi-objective operation of cascade reservoirs.

Acknowledgment

This study is financially supported by the National Key Research and Development Program of China (2016YFC0402206 and 2018YFC0407904), the Research Council of Norway (FRINATEK Project 274310) and the Ministry of Science and Technology, Taiwan (MOST 106-2627-M-002 -025, 107-2621-M-002 -004 -MY3). The datasets provided by the Changjiang Water Resources Commission in China are acknowledged. The authors would like to thank the Editors and anonymous Reviewers for their constructive comments that greatly contributed to improving the manuscript.

References

Allawi, M. F., Jaafar, O., Hamzah, F. M., & El-Shafie, A., 2019. Novel reservoir system 821 822 simulation procedure for gap minimization between water supply and demand. J. of Clean. Prod. 206, 928-943. 823

1

2

3

5

б

7 8

9

10

13

16

17

20

21

23

24

25

27

28

30

31

32

34

35

38

39

42

43

45

46

47

61 62

- Bai, T., Kan, Y.B., Chang, J.X., Huang, Q., Chang, F.J., 2017, Fusing feasible search space 4 824 into PSO for multi-objective cascade reservoir optimization, Appl. Soft. Comput. 51, 825 826 328-340.
- 827 Bianchi L, Gambardella LM, Dorigo M., 2002. An ant colony optimization approach to the probabilistic traveling salesman problem. In: Proceedings of the 7th international 828 conference on parallel problem solving from nature. New York: Springer-Verlag, 883-11 829 12 830 892.
- Bora, T. C., Coelho, L. D. S., & Lebensztajn, L., 2012. Bat-inspired optimization approach for 14 831 15 **832** the brushless DC wheel motor problem. IEEE Trans. Magn. 48(2), 947-950.
- 833 Cheng, C. T., Shen, J. J., Wu, X. Y., & Chau, K. W., 2012. Operation challenges for fast-18 834 growing China's hydropower systems and respondence to energy saving and emission ¹⁹ 835 reduction. Renewable Sustainable Energy Rev. 16(5), 2386-2393.
- Cheng, C. T., Chen, F., Li, G., Ristić, B., Mirchi, A., Qiyu, T., & Madani, K., 2018. Reform 836 and renewables in China: The architecture of Yunnan's hydropower dominated 22 **837** 838 electricity market. Renewable Sustainable Energy Rev. 94, 682-693.
- Deb, K., Pratap, A., Agarwal, S., & Meyarivan, T. A. M. T., 2002. A fast and elitist 839 26 840 multiobjective genetic algorithm: NSGA-II. IEEE Trans. Evolu. Comput. 6(2), 182-197. 841
- Dou, X., 2013. Low carbon-economy development: China's pattern and policy selection. 29 **842** Energy Policy. 63, 1013-1020. 843
- Ehteram, M., Karami, H., Mousavi, S. F., Farzin, S., & Kisi, O., 2017. Optimization of energy 844 management and conversion in the multi-reservoir systems based on evolutionary 33 **845** 846 algorithms. J. of Clean. Prod. 168, 1132-1142.
- ₃₆ 847 Ehteram, M., Karami, H., & Farzin, S., 2018a. Reservoir optimization for energy production 37 using a new evolutionary algorithm based on multi-criteria decision-making models. 848 Water Resour. Manage. 32(7), 2539-2560. 849
- Ehteram, M., Karami, H., Mousavi, S. F., Farzin, S., Celeste, A. B., & Shafie, A. E. (2018b). 40 850 ⁴¹ 851 Reservoir Operation by a new evolutionary algorithm: Kidney Algorithm. Water 852 Resour. Manage. 32(14), 4681-4706.
- Ekinci, S., Demiroren, A., & Hekimoglu, B., 2018. Parameter optimization of power system 44 853 stabilizers via kidney-inspired algorithm. Transactions of the Institute of Measurement 854 and Control, 1112-1120. 855
- ⁴⁸ 856 Feng, Z. K., Niu, W. J., & Cheng, C. T., 2018a. Optimizing electrical power production of 49 hydropower system by uniform progressive optimality algorithm based on two-stage 857 50 51 858 search mechanism and uniform design. J. of Clean. Prod. 190, 432-442.
- 52 859 Feng, Z. K., Niu, W. J., Cheng, C. T., & Wu, X. Y., 2018b. Optimization of large-scale 53 hydropower system peak operation with hybrid dynamic programming and domain 860 54 knowledge. J. of Clean. Prod. 171, 390-402. 55 **861**
- 56 Fister Jr, I., Yang, X. S., Fister, I., Brest, J., & Fister, D., 2013. A brief review of nature-862 57 inspired algorithms for optimization. Elektrotehniski Vestnik, 80, 1-7. 58 **863**
- ⁵⁹ 864 Geem, Z. W., Kim, J. H., & Loganathan, G. V., 2001. A new heuristic optimization algorithm: 60
 - 44

- 865 harmony search. Simulation. 76(2), 60-68.
- 866 Goldberg D. E., 1989. Genetic Algorithms in search, optimization and machine learning, New York, Addison-Wesley. 867
- Hashimoto, T., Stedinger, J. R., & Loucks, D. P., 1982. Reliability, resiliency, and 4 868 vulnerability criteria for water resource system performance evaluation. Water Resour. 869 Res. 18(1), 14-20. 870
- He, S., Guo, S., Chen, K., Deng, L., Liao, Z., Xiong, F., & Yin, J., 2019. Optimal 871 impoundment operation for cascade reservoirs coupling parallel dynamic 872 programming with importance sampling and successive approximation. Adv. Water 11 873 Resour. 131, 103375. 874
- He, S., Yin, X. A., Yu, C., Xu, Z., & Yang, Z., 2018. Quantifying parameter uncertainty in ₁₄ 875 15 876 reservoir operation associated with environmental flow management. J. of Clean. Prod. 877 176, 1271-1282.
- Hatamlou, A., 2013. Black hole: A new heuristic optimization approach for data clustering. 18 **878** ¹⁹ 879 Inf. Sci. 222, 175-184.
- 21 **880** Hosseini, H. S., 2007. Problem solving by intelligent water drops. IEEE congress on evolutionary computation, 3226-3231. 22 **881**
- ²³ 882 Hu, Z., Yuan, J., & Hu, Z., 2011. Study on China's low carbon development in an Economy-25 **883** Energy-Electricity-Environment framework. Energy Policy. 39(5), 2596-2605.
- 26 884 Jha, S. K., Bilalovic, J., Jha, A., Patel, N., & Zhang, H., 2017. Renewable energy: Present research and future scope of Artificial Intelligence. Renewable Sustainable Energy Rev. 885 29 **886** 77, 297-317.
- ³⁰ 887 Jaddi, N. S., Alvankarian, J., & Abdullah, S., 2017. Kidney-inspired algorithm for optimization problems. Communications in Nonlinear Science and Numerical 888 Simulation, 42, 358-369. 33 **889**
- 890 Jaddi, N. S., & Abdullah, S. (2018). Optimization of neural network using kidney-inspired ₃₆ 891 algorithm with control of filtration rate and chaotic map for real-world rainfall forecasting. Engineering Applications of Artificial Intelligence, 67, 246-259. 37 892
- Johnson, D. S., Aragon, C. R., McGeoch, L. A., & Schevon, C., 1989. Optimization by 893 simulated annealing: an experimental evaluation; part I, graph partitioning. Oper. Res. 40 894 ⁴¹ 895 37(6), 865-892.
- 896 Kennedy J, Eberhart R., 1995. Particle swarm optimization. In: Neural networks, proceedings, IEEE international conference, 1942-1950. 44 897
- 45 Maarouf, M., Sosa, A., Galván, B., Greiner, D., Winter, G., Mendez, M., & Aguasca, R., 2015. 898 46 The role of artificial neural networks in evolutionary optimisation: a review. In 47 **899** 48 900 Advances in evolutionary and deterministic methods for design, optimization and 49 901 control in engineering and sciences Springer, Cham, 59-76. 50
- Molina, D., LaTorre, A., & Herrera, F., 2018. An insight into bio-inspired and evolutionary 51 **902** ⁵² 903 algorithms for global optimization: review, analysis, and lessons learnt over a decade of competitions. Cog. Comput. 10, 517-544. 904
- Ming, B., Liu, P., Guo, S., Cheng, L., Zhou, Y., Gao, S., & Li, H., 2018a. Robust hydroelectric 55 **905** unit commitment considering integration of large-scale photovoltaic power: A case 906 58 **907** study in China. Appl. Energy. 228, 1341-1352.
- 59 908 Ming, B., Liu, P., Cheng, L., Zhou, Y., & Wang, X., 2018b. Optimal daily generation

- 62 63
 - 64 65

1

2

3

5

б

7 8

9

10

12

13

16

17

20

24

27

28

31

32

34

35

38

39

42

43

53

54

56

57

- 909 scheduling of large hydro-photovoltaic hybrid power plants. Energy Convers. Manage. 910 171, 528-540.
- 2 Pham DT, Ghanbarzadeh A, KocE, Otri S, Rahim S, Zaidi M., 2005. The bees algorithm. 911 3 Cardiff University, UK, Technical Note, Manufacturing Engineering Centre, Cardiff. 4 912
- 5 REN21, R. Renewables 2017: global status report, REN21 secretariat, Paris, France 2017. In: 913 б 7 914 Tech Rep. <www.ren21.net/wp-content/uploads/2017/06/17-8399_GSR_2017_Full_ 8 Report_0621_Opt.pdf.> 915
- 9 Shen, J., Zhang, X., Wang, J., Cao, R., Wang, S., & Zhang, J., 2019. Optimal operation of 916 10 interprovincial hydropower system including Xiluodu and local plants in multiple 11 917 ¹² 918 recipient regions. Energies. 12(1), 144. 13
- ₁₄ 919 Singh, V. K., & Singal, S. K., 2017. Operation of hydro power plants-a review. Renewable 15 920 Sustainable Energy Rev. 69, 610-619.
- 16 921 Srinivas, M., & Patnaik, L. M., 1994. Adaptive probabilities of crossover and mutation in 17 18 **922** genetic algorithms. IEEE Transactions on Systems, Man, and Cybernetics, 24(4), 656-¹⁹ 923 667.
- 20 Uen, T. S., Chang, F. J., Zhou, Y., & Tsai, W. P., 2018. Exploring synergistic benefits of 924 21 Water-Food-Energy Nexus through multi-objective reservoir optimization schemes. 22 **925** ²³ 926 Sci. Total Environ. 633, 341-351. 24
- 25 **927** Wang, J., Chen, C., & Liu, S., 2018. A new field-levelling procedure to minimize spillages in 26 928 hydropower reservoir operation. Energy. 160, 979-985.
- Wardlaw, R., & Sharif, M., 1999. Evaluation of genetic algorithms for optimal reservoir 929 system operation. J. Water Res. Plan. Manage. 125(1), 25-33. 29 **930**
- ³⁰ 931 Yang, X. S., & Deb, S., 2009. Cuckoo search via Lévy flights. In World Congress on Nature & Biologically Inspired Computing (NaBIC), IEEE, 210-214. 932
- Yang, X. S., 2010a. A new metaheuristic bat-inspired algorithm. In Nature inspired 33 **933** ³⁴ 934 cooperative strategies for optimization. Springer, Berlin, Heidelberg, 65-74. 35
- ₃₆ 935 Yang, X. S., 2010b. Firefly algorithm, Levy flights and global optimization. In Research and development in intelligent systems XXVI. Springer, London, 209-218. 37 936 38
- Zhang, J., Chung, H. S. H., & Lo, W. L., 2007. Clustering-based adaptive crossover and 937 mutation probabilities for genetic algorithms. IEEE Transactions on Evolutionary 40 938 ⁴¹ 939 Computation, 11(3), 326-335.
- 42 940 Zhou, Y., Guo, S., Liu, P., & Xu, C., 2014. Joint operation and dynamic control of flood 43 limiting water levels for mixed cascade reservoir systems. J. Hydro. 519, 248-257. 44 941
- 45 Zhou, Y., Guo, S., Xu, C. Y., Liu, P., & Qin, H., 2015. Deriving joint optimal refill rules for 942 46 ₄₇ 943 cascade reservoirs with multi-objective evaluation. J. Hydro. 524, 166-181.
- 48 944 Zhou, Y., Guo, S., Hong, X., & Chang, F. J., 2017. Systematic impact assessment on inter-49 basin water transfer projects of the Hanjiang River Basin in China. J. Hydro. 553, 584-945 50 51 946 595.
- ⁵² 947 Zhou, Y., Guo, S., Chang, F. J., Liu, P., & Chen, A. B., 2018a. Methodology that improves 53 water utilization and hydropower generation without increasing flood risk in mega 948 54 cascade reservoirs. Energy. 143, 785-796. 55 **949**
- 56 Zhou, Y., Guo, S., Chang, F. J., & Xu, C. Y., 2018b. Boosting hydropower output of mega 950 57 58 **951** cascade reservoirs using an evolutionary algorithm with successive approximation. 59 **952** Appl. Energy. 228, 1726-1739.
- 60 61 62

27

28

31

32

- 63 64
- 65

Stimulate hydropower output of mega cascade reservoirs using an improved Kidney

Algorithm

Yanlai Zhou^{1,2}, Shenglian Guo^{1,*}, Chong-Yu Xu^{1,2,*}, Fi-John Chang³, Hua Chen¹, Pan Liu¹, Bo Ming¹

¹ State Key Laboratory of Water Resources and Hydropower Engineering Science, Wuhan University, Wuhan, 430072, China.

² Department of Geosciences, University of Oslo, P.O. Box 1047 Blindern, N-0316 Oslo, Norway.

³ Department of Bioenvironmental Systems Engineering, National Taiwan University, Taipei, 10617, Taiwan.

* Correspondence to: Shenglian Guo (slguo@whu.edu.cn); Chong-Yu Xu (c.y.xu@geo.uio.no).